Accelerator Status Report

Run Schedule: Aug 21, 2009 to Dec 22, 2009
Dec. 16, 2009

Injector

· Injector was stable. 

· (12/11) Orbits were adjusted to stabilize Hall B BOM counts.

· (12/09) After the spot move the orbits were adjusted a bit for high current run

· (12/09) The laser spot was moved for better QE.
· The vertical polarization run went well and the setup was backed out. Then the spin dance was performed successfully.

· We are planning to have a spot move today. 

Accelerator   Energy: 6.068 GeV
· Overall, the accelerator was running well with high current and low trip rates.
· RF: with high current to Hall A there are some unstable cavities. We need lower the gsets of unstable cavities. 

· (12/16) CMAC power supply of zone NL14 failed. (4.5 h)

· (12/12) RSEP9A power supply failed. (14.0 h)
· VBV2L17A and VBV2L17B would open and the faulty parts were replaced. (6.0 h)
· 12/9 MAN5R04 kept tripping off. The root cause was the building temperature was too high. (5.5 h) 
· The accelerator has been running well with high current (115uA) to Hall A and low trip rates. The optics has been stable. The path length has been slowly drifting in a way that Hall A Compton rates are stable and low, and Hall B BOM rates are low.

· The tails on 1A and 2A SLMs are taken out and the path length was adjusted to compensate for the change. 

· (12/4) VBV1L06A position indicator failed and the faulty part was replaced.

· (12/9) MAN5R04 had problems again. It kept tripping off. EES would replace the shunt 
module.
Hall A---PVDIS; pass 5; energy 6.068 GeV
· Beam was good.

· Hall A Compton counts are good and stable.

· Hall A electron counts are high. Experts spent some time trying to lower it but there are still some issues with it. 

Hall B ---Eg6; pass 5; energy 6.068 GeV
· Beam was good.

· During the vertical polarization run Hall B was changed to pass 1. At 1 pass Hall B was not happy with its BOM rates (too high) and beam profile (too large). Later the hall decided to back out the pass change without taking data. Please call for help if you run into problems. Hall B will try pass 1 again. 
Hall C---HKS; pass 5; energy 6.068 GeV
· Hall C is down for installation of Qweak. It will be back at the end of May in 2010.
· HKS ended successfully on Nov. 2.

